Author's response to reviews

Title: Tumour stromal cells derived from paediatric malignancies display MSC-like properties and impair NK cell cytotoxicity

Authors:

Pascal-David Johann (pascal-david.johann@student.uni-tuebingen.de)
Martin Vaegler (martin.vaegler@med.uni-tuebingen.de)
Friederike Gieseke (friederike.gieseke@med.uni-tuebingen.de)
Philippa Mang (Philippa.Mang@med.uni-tuebingen.de)
Sorin Armeanu-Ebinger (sorin.armeanu-ebinger@med.uni-tuebingen.de)
Torsten Kluba (Torsten.Kluba@med.uni-tuebingen.de)
Rupert Handgretinger (Rupert.Handgretinger@med.uni-tuebingen.de)
Ingo Müller (Ingo.Mueller@med.uni-tuebingen.de)

Version: 7 Date: 21 September 2010

Author's response to reviews:

I changed the FAX number.
While you were editing figure 2, part a) changed into something like a photo of a stary night. Please revert to the earlier version.